Iowa State University 1 CprE 489, Lecture 6

CprE 489 Lecture 6

Fall Semester 1997

Iowa State University 2 CprE 489, Lecture 6

Multiplexing

• Multiplexing is using a single carrier for multiple channels

• Frequency Division Multiplexing (FDM)- Each channel uses a different frequency band.

• Time Division Multiplexing (TDM)- Each channel uses a different time slot in a round robin fashion • Both FDM and TDM may not utilize the carrier well. Neither is appropriate for data communications.

a a a b b b c c c

Iowa State University 3 CprE 489, Lecture 6

Asynchronous Time Division Multiplexing

• Like TDM but time slots are not allocated in round robin fashion.

• For an ATDM to work there must be a policy for managing the incoming data.

a a a b b b c c c a b c - - c a - c TDM

a a a b b b c c c

ATDM a a c b b a a a

Iowa State University 4 CprE 489, Lecture 6

Issues in Multiplexing

• Quality of service

- Do the incoming channels receive their fair share of the outgoing bandwidth? - Is the data from one channel unduly delayed?

• Bandwidth allocation - • Buffer control -